Author's response to reviews

Title: Significant cross reactive antibodies to influenza virus in adults and children during a period of marked antigenic drift

Authors:

Michal Mandelboim (michal.mandelboim@sheba.health.gov.il)
Michal Bromberg (michal.Bromberg@ICDC.HEALTH.GOV.IL)
Hilda Sherbany (hilda.sherbany@sheba.health.gov.il)
Inbar Zucker (Inbar.Zucker@ICDC.HEALTH.GOV.IL)
Karnit Yaary (karnitina@walla.com)
Ravit Bassal (Ravit.Bassal@ICDC.HEALTH.GOV.IL)
Rita Dichtiar (Rita.Dichtiar@ICDC.HEALTH.GOV.IL)
Danny Cohen (dancohen@post.tau.ac.il)
Tamar Shohat (Tamar.Shohat@ICDC.HEALTH.GOV.IL)
Ella Mendelson (ella.mendelson@gmail.com)
Manfred Green (manfred.s.green@gmail.com)

Version: 5 Date: 25 May 2014

Author's response to reviews: see over
1.5.2014

Dear Editor,

Thank you for your email. We revised our manuscript in complete accordance with the editor request and the manuscript was also edited by a native English speaker as requested (changes are marked in red). Below please find our point by point response to the editor. I hope that you will now accept this manuscript for publication.

Best wishes,
Michal

Comment: 1. Page 5, line 97, suggest to add a few words to make it more clear: "The children analyzed in this study were not vaccinated against influenza by the time of recruitment."
Response: Corrected

Comment: 2. Page 7, line 145, duplicated "in which".
Response: Corrected

Comment: 3. Figure 1, move one slash after "7" for virus A/California/7/2004.
Response: Corrected

Comment: 4. Page 8, line 153, suggest to reword the first sentence of highlighted content as "Importantly, it's necessary to clarify whether the children that evaluated in this manuscript were vaccinated against influenza."
Response: Corrected

Editorial Requirement:
Copyediting:

Comment: We recommend that you ask a native English speaking colleague to help you copyedit the paper. If this is not possible, you may need to use a professional language editing service.
Response: The manuscript was edited by a native English speaker.